� FDM: Transmit each flow over the physical link at a different frequency band � Example: TV channels

Amplitude

Frequency

Time

1 2 3 4 5 6 7 8 9 1010 1111 1212 f1f1 f2f2 f3f3 f4f4 f5f5

Frequency Division Multiplexing

� TDM: Transmit each flow over the physical link at a different time slot - round robin

Amplitude

Frequency

Time

1 2 3 4 5 6 7 8 9 1010 1111 1212 f1f1 f2f2

Time Division Multiplexing

� ATDM: Transmit each flow over the physical link at a different time slot - however not round robin, but according to a pre-established policy which better utilizes the channel bandwidth.

Amplitude

Frequency

Time

1 2 3 4 5 6 7 8 9 1010 1111 1212 f1f1 f2f2

Async. Time Division Multiplexing